Author's response to reviews

Title: Awareness, attitudes, and practices related to the swine influenza pandemic among the Saudi public

Authors:

Hanan H Balkhy (Balkhyh@ngha.med.sa)
Mostafa A Abolfotouh (mabolfotouh@gmail.com)
Rawabi H Al-Hathlool (r.alhathlool@gmail.com)
Mohammed A Al-Jumah (Jumahm@ngha.med.sa)

Version: 5 Date: 23 February 2010

Author's response to reviews:

Dear Editor,

Thank you for the gally proof of the manuscript.

Just few modifications have been done in this resubmitted revised version. These modifications are:

1. the e-mail of first author, balkhyh@ngha.med.sa was changed to balkhyh@hotmail.com

2. word "extreamly" in page 6, paragraph 2 was corrected to "extremely".

3. Extra spaces between words were deleted as shown in the text Comment boxes.

4. paragraphs 3&4 in page 11 were reformatted in a double space format.

5. In conclusion section, page 12, last paragraph, the sentence "and this finding may be related to the public's questioning of reports by the Ministry of Health" was deleted.

Regards

Mostafa Abolfotouh